Author's response to reviews

Title: Methylene Blue-Assisted Technique for Harvesting Lymph Nodes after Radical Surgery for Gastric cancer: A Prospective Randomized Phase III Study

Authors:

Toru Aoyama (t-aoyama@lilac.plala.or.jp)
Takaki Yoshikawa (yoshikawat@kcch.jp)
Satoshi Morita (smorita@urate.yokohama-cu.ac.jp)
Junya Shirai (gratefanta1123@yahoo.co.jp)
Hirohito Fujikawa (hfujikawa@kcch.jp)
Kenichi Iwasaki (kiwasaki@kcch.jp)
Tsutomu Hayashi (hayashit@kcch.jp)
Takashi Ogata (ogatat@kcch.jp)
Haruhiko Cho (choharuhiko@kcch.jp)
Norio Yukawa (nryukawa@mac.com)
Takashi Oshima (ohshimatakashi@yahoo.co.jp)
Yasushi Rino (rino@med.yokohama-cu.ac.jp)
Munetaka Masuda (mmasuda@yokohama-cu.ac.jp)
Akira Tsuburaya (tuburayaa@kcch.jp)

Version: 3 Date: 18 November 2013

Author's response to reviews: see over
Dear editor,

Thank you for your prompt review and valuable, constructive comments for our manuscript. In respond to the comments, we prepared the answers and revised the manuscript.

Thank you again for your kind consideration, and we are looking forward to your reply. Respectfully yours

Takaki Yoshikawa (Corresponding author)
Department of Gastrointestinal Surgery,
Kanagawa Cancer Center
1-1-2 Nakao, Asahi-Ku, Yokohama 241-0815, Japan
Tel:+81-45-391-5761
Fax:+81-45-361-4692
E-mail: yoshikawat@kcch.jp